Author’s response to reviews

Title: Male Partner Antenatal Clinic Attendance is Associated with Increased Uptake of Maternal Health Services and Infant BCG Immunization A National Survey in Kenya

Authors:

Beryne Odeny (bodeny@uw.edu)
McGrath Christine (mcgrathc@uw.edu)
Langat Agnes (vpl4@cdc.gov)
Pintye Jillian (jpintye@u.washington.edu)
Singa Benson (singabo2008@gmail.com)
Kinuthia John (kinuthia@uw.edu)
Katana Abraham (wvf0@cdc.gov)
Nganga Lucy (hon5@cdc.gov)
John-Stewart Grace (gjohn@uw.edu)

Version: 2 Date: 26 Jul 2019

Author’s response to reviews:

07/22/2019

Beryne Odeny

To: Editorial team, BMC Pregnancy and Childbirth


We thank the reviewers for their helpful input. Please see responses to the reviewer comments below. The responses have been typed in bold.
1. Please state in the cover letter whether the maps depicted in figure 1 are your own or taken from another source.

RESPONSE:

Thank you for pointing this out. This map was generated by one of the co-authors, Jillian Pintye, who has granted permission to use in this manuscript. This acknowledgement is written in the figure legend.

2. Please include the full name of each of the ethics committees that had approved your study, in the Ethics approval and consent to participate section.

RESPONSE:

Thank you noting this. The full names of each of the ethics committees have been included as requested. It now reads:

“Ethical approvals were obtained from the University of Washington Human Subjects Division, the Kenya Medical Research Institute Ethical Review Committee, and the Associate Director for Science at the U.S. Centers for Disease Control and Prevention.”

3. In the Funding section, please also state the role of the funding body in the design of the study, collection, analysis, interpretation of data, and in writing the manuscript.

RESPONSE:

Thank you for this comment. We have stated the role of the funding body as follows:

“The funders participated in design of the study, collection and interpretation of the data, and in review of this manuscript.”

4. Please remove the authorship disclaimer that is included in the Declarations section.

RESPONSE:

This has been removed as recommended.

5. We note that many of the initials included in the Authors' Contributions section do not match the names included in the list of authors. As such, please ensure that each authors initial is
included in the order of the authors first name (given name) and last name (family name). For example: Beryne Odeny should be included as BO.

RESPONSE:

Thank you for this note. Authors’ initials in the Authors’ Contributions section have been updated per the journal’s stipulations and now match the names in the list of authors.

6. Please proofread and ensure that when you upload your revised submission that it is in the final form for publication. Please remove any tracked changes colored text or highlighting and include only a single clean copy of the manuscript. Should you wish to respond to these revision requests, please include the information in the designated input box only.

RESPONSE:

Thank you for your review. A clean copy of the manuscript has been uploaded.

7. Please complete all the declarations details.

- Availability of data and materials.

RESPONSE:

Thank you for your comment. The Declarations’ section on 'Availability of data and materials' has been completed as requested.